\begin{table}%t4
\caption{Foreground intermediate-type stars and lightly reddened,
unclassified stars.}\label{tab_foreground_oth}
%\centerline
 {\begin{tabular}{ccccccll}
\hline\hline
Star & $\alpha (2000)$ & $\delta(2000)$ & $J$ & $H$ & $K_{\rm S}$ & Other IDs & Notes \\
\hline \noalign{ {\it Foreground stars} }\\
  28 & 20  29  28.37 &  42  19  00.3 &  6.814 & 6.579 & 6.521 & \object{HD~195405} & 1 \\
  35 & 20  38  13.10 &  39  32  49.5 &  6.941 & 6.705 & 6.660 & \object{HD~196790}& 2\\
  57 & 20  24  36.12 &  41  30  02.8 &  7.603 & 7.281 & 7.192 & \object{LTT~15972} & 3 \\
  58 & 20  37  29.72 &  42  33  14.9 &  7.651 & 7.305 & 7.219 & \object{HIP~101763} & 4 \\
  69 & 20  25  39.03 &  41  53  45.6 &  7.673 & 7.428 & 7.370 & \object{HD~229293} & 5 \\
  73 & 20  24  03.81 &  40  20  22.7 &  7.653 & 7.454 & 7.391 & \object{HD~229244} & 6 \\
  85 & 20  29  46.12 &  39  39  01.1 &  7.842 & 7.572 & 7.516 & \object{BD+39~4212} & 7 \\
  92 & 20  39  34.11 &  42  01  30.3 &  7.784 & 7.670 & 7.616 & \object{BD+41~3844} & 8 \\
 174 & 20  29  52.91 &  42  21  27.4 &  8.714 & 8.481 & 8.423 & \object{BD+41 3776} & 9 \\
 203 & 20  40  07.90 &  41  15  10.8 &  9.067 & 8.708 & 8.615 & \object{BD+40~4272} & 10 \\
\noalign{ {\it Lightly reddened stars with no classification
in the visible}
}\\
 139 & 20  35  35.00 &  40  01  29.5 &  8.380 & 8.211 & 8.128 & & \\
 191 & 20  38  46.80 &  42  01  06.4 &  8.908 & 8.615 & 8.552 & & \\
 197 & 20  26  36.09 &  40  28  29.1 &  8.810 & 8.606 & 8.583 & & \\
 201 & 20  24  22.65 &  41  26  08.5 &  8.901 & 8.652 & 8.612 & \object{HD~229266} & \\
 210 & 20  24  16.06 &  40  29  26.6 &  8.964 & 8.686 & 8.642 & \object{HD~229252} & \\
 214 & 20  39  28.12 &  40  36  15.1 &  9.020 & 8.747 & 8.677 & \object{BD+40~4266} & \\
 216 & 20  24  25.50 &  39  49  28.4 &  8.831 & 8.748 & 8.695 & \object{HD~229258} & \\
 231 & 20  30  02.17 &  42  14  34.1 &  9.053 & 8.845 & 8.767 & & \\
 234 & 20  23  59.70 &  40  07  46.8 &  8.952 & 8.820 & 8.783 & \object{HD~229237} & \\
 239 & 20  39  10.53 &  40  20  06.4 &  9.149 & 8.861 & 8.813 & & \\
 249 & 20  27  52.05 &  41  31  20.0 &  9.143 & 8.895 & 8.847 & & \\
\hline 
\end{tabular}}
\smallskip
 Notes\\
  1: Classified as G0Iab: (Nassau \& Morgan~\cite{nassau52}) and as
G2IV (Bidelman~\cite{bidelman57}).\\
2: Classified as F8IV (Nordstr\o m et~al.~\cite{nordstroem04}).\\
3: High proper motion, horizontal branch star; see e.g.
Behr~(\cite{behr03}).\\
4: Close binary, high proper motion star; Fabricius \&
Makarov~(\cite{fabricius00}).\\
5: Classified as K0 in the HD catalog.\\
6: Classified as G0 in the HD catalog.\\
7: Classified as F8 in the SAO catalog.\\
8: Classified as F0V by Straizys et~al.~(\cite{straizys89}) based
on Vilnius photometry.\\
9: Classified as F5:V by Straizys et~al.~(\cite{straizys89}) based
on Vilnius photometry.\\
10: Nearby, high proper motion star at 75~pc according to the
Hipparcos catalog. Suspected metal-poor subdwarf by Reid et~al.~(\cite{reid01}).\\
\end{table}